Author's response to reviews

Title: Oral Squamous Cell Carcinoma Proliferative Phenotype is Modulated by Proanthocyanidins: A Potential Prevention and Treatment Alternative for Oral Cancer

Authors:

Michael King (michael.king@sdm.unlv.edu)
Kourt Chatelain (kourt.chatelain@sdm.unlv.edu)
Dustin Farris (dustin.farris@sdm.unlv.edu)
Dayne Jensen (dayne.jensen@sdm.unlv.edu)
Jason Pickup (jason.pickup@sdm.unlv.edu)
Aaron Swapp (aaron.swapp@sdm.unlv.edu)
Susan O'Malley (susan.omalley@unlv.edu)
Karl Kingsley (karl.kingsley@unlv.edu)

Version: 2 Date: 14 February 2007

Author's response to reviews: see over
A minor problem resulted in a formatting error, which has been corrected. Please accept our apologies. Many thanks - KK